Author’s response to reviews

Title: Nurse clinic versus home delivery of evidence-based community leg ulcer care: A randomized health services trial

Authors:

Margaret B Harrison (Margaret.B.Harrison@Queensu.ca)
Ian D Graham (igraham@cihr-irsc.gc.ca)
Karen Lorimer (klorimer@carefor.ca)
Elizabeth VandenKerkof (EV5@Queensu.ca)
Maureen Buchanan (Baybuch@kos.net)
Phil S Wells (pwells@ottawahospital.on.ca)
Tim Brandys (TBRANDYS@ottawahospital.on.ca)
Tadeusz Pierscianowski (tal@magma.ca)

Version: 8 Date: 14 November 2008

Author’s response to reviews:

Affiliations with a country for each address - now complete
apostrophe added -> Authors’ contributions, as requested
Additional files = none; tables included at the end of the manuscript, as suggested